Author's response to reviews

Title: Diabetes in Sub Saharan Africa 1999-2010: Epidemiology and Public Health Implications

Authors:

Victoria J Hall (victoriajhall@yahoo.co.uk)
Reimar W Thomsen (r.thomsen@rn.dk)
Ole Henriksen (ohrk@novonordisk.com)
Nicolai Lohse (nilh@novonordisk.com)

Version: 2 Date: 17 November 2010

Author's response to reviews:

Dear Editors,

I have made one change to the manuscript, which is to change the title from "Diabetes in Sub-Saharan Africa 1999-2010: Epidemiology and Public Health Implications" to "Diabetes in Sub-Saharan Africa 1999-2010: Epidemiology and Public Health Implications. A systematic review."

I have also uploaded a Prisma checklist. I have kept the flow diagram (Annex 1)describing the review, which follows the format of the flow diagram on the Prisma website.

I look forward to hearing from you.

Yours sincerely,

Victoria Hall